Author's response to reviews

Title: Gastric lipoma presenting as a giant bulging mass in an oligosymptomatic patient: a case report

Authors:

Francisco AF Neto (estertores@hotmail.com)
Maria CF Ferreira (cfurian@terra.com.br)
Luiz CN Bertoncello (luiz-bertoncello@uol.com.br)
Andre A Neto (andreamateneto@yahoo.com.br)
Wilson C Aveiro (marcopacifier@gmail.com)
Caroline A Bento (agnelli.carol@gmail.com)
Gustavo N Cecchino (gusta.nardini@gmail.com)
Marco AM Rocha (marcoantonio@rock.com)

Version: 4 Date: 15 August 2012

Author's response to reviews:

The Deputy Editor’s concern has been addressed. The final language revision is highlighted in red. If there are any other corrections we kindly ask you to be more specific. Thanks